Author's response to reviews

Title: Predicting Invasive Breast Cancer versus DCIS in Different Age Groups

Authors:

Mehmet U S Ayvaci (mua120030@utdallas.edu)
Oguzhan Alagoz (alagoz@engr.wisc.edu)
Jagpreet Chhatwal (chhatwal@pitt.edu)
Alejandro Munoz del Rio (amunozdelrio@uwhealth.org)
Edward Sickles (edward.sickles@ucsfmedctr.org)
Houssam Nassif (nassif@wisc.edu)
Karla Kerlikowske (karla.kerlikowske@ucsf.edu)
Elizabeth S Burnside (eburnside@uwhealth.org)

Version: 2 Date: 11 June 2013

Author's response to reviews: see over
June 11, 2013

To Whom It May Concern:

This letter accompanies a manuscript entitled “Predicting Invasive Breast Cancer versus DCIS in Different Age Groups” which I am submitting as a research article. This manuscript went through the initial review process at the BMC Cancer and advised to be resubmitted with some formatting changes. Below, we explain the changes made to the manuscript requested by the editorial office:

1) The samples were collected at the University of California, San Francisco and we updated the abstract and the body of the manuscript with this information.
2) Data was collected prospectively and this information is incorporated into the abstract and the body of the manuscript.

Please send all correspondence concerning the manuscript to me at the address listed on the title page. Thank you for your assistance.

Sincerely,

Elizabeth S. Burnside, M.D., M.P.H., M.S.
Department of Radiology, University of Wisconsin Medical School
E3/311 CSC, 600 Highland Ave, Madison, WI 53792-3252
Phone: (608) 265-2021 Fax: (608) 265-1836
Email: es.burnside@hosp.wisc.edu